Author's response to reviews

Title: Recruitment of young women to a trial of chlamydia screening - as easy as it sounds?

Authors:

Helen Atherton (hatherto@sgul.ac.uk)
Debbie Banks (deborahbanks@gmail.com)
Ruth Harbit (ruthharbit@cantab.net)
Linzie Long (m0500727@sgul.ac.uk)
Fiona Chadd (m0500446@sgul.ac.uk)
Phillip Hay (phay@sgul.ac.uk)
Sally Kerry (sgjdgp@sgul.ac.uk)
Ian Simms (ian.simms@hpa.org.uk)
Pippa Oakeshott (oakeshot@sgul.ac.uk)

Version: 5 Date: 2 December 2007

Author's response to reviews: see over
The Editor
Trials
Biomed Central
02/12/2007

MS: 1347106327144745- Recruitment of young women to a trial of chlamydia screening - as easy as it sounds?

Dear Editor

Here are detailed the further formatting changes as requested.

1) References have been cited in the text using consecutive numbers in brackets.
2) ‘Authors’ Contributions’ has been amended accordingly.
3) Competing interests has been changed to sentence case and moved before authors’ contributions.
4) Funding has now been incorporated as part of the acknowledgements section.

We very much hope these additional amendments are acceptable.

With Best Wishes

Yours sincerely

Helen Atherton